Author's response to reviews

Title: Alpha vermiform appendix presenting in an inguinal hernia - Amyand's hernia: a case series

Authors:

Kyriakos Psarras (psarras_k@yahoo.com)
Miltiadis A Lalountas (miltiadislalountas@yahoo.gr)
Minas Baltatzis (minbalt@yahoo.gr)
Efstathios Pavlidis (pavlidakos@hotmail.com)
Anastasios Tsitlakidis (skrits7@hotmail.com)
Nikolaos Symeonidis (niksym@hotmail.com)
Konstantinos Ballas (mpallas@auth.gr)
Theodoros Pavlidis (pavlidth@med.auth.gr)
Athanassios Sakantamis (asakantam@med.auth.gr)

Version: 9 Date: 5 August 2011

Author's response to reviews: see over
Reply to the Editors’ comments

With respect to the Editors’ comments, we made all the necessary changes in this revised version of the manuscript.

1. We amend the title in a more descriptive, as follows: A vermiform appendix presenting in an inguinal hernia - Amyand's hernia: a case series.

2. We state individually the age, sex and ethnicity for each of the four patients in the abstract (case presentation).

The authors would like to thank the Editors for their cooperation.

Miltiadis Lalountas,
Surgeon MD, MSc, PhD Candidate